Author's response to reviews

Title: Effects of a community-based healthy heart program on increasing healthy women's physical activity: a randomized controlled trial guided by Community-based Participatory Research (CBPR)

Authors:

Iraj Nabipour (nabipourpg@bpums.ac.ir)
Raha Pazoki (pazkipg@bpums.ac.ir)
Nasrin Seyednezami (Nasrin_naz2001@yahoo.com)
Seyed Reza Imami (inabim@yahoo.com)

Version: 2 Date: 18 December 2006

Author's response to reviews:

Dear Chantal botha,
I'm writing this letter to inform you about the changes has been made in the revised manuscript of the article number 1363859238124560 entitled: "Effects of a community-based healthy heart program on increasing women's physical activity: a randomized controlled trial guided by Community-based Participatory Research (CBPR)" Submitted on 6 Dec 2006.
Upon your notification, the registration number has been added at the last part of the abstract, and the phrase of "women's physical activity" has been replaced by "healthy women's physical activity". Names of two first authors have been replaced based on agreement of authors. In addition, list of abbreviation has been added to the last part of the article.

With the best regards

Raha Pazoki
Iraj Nabipour
Nasrin Seyednezami
Seyed Reza Imami